Author's response to reviews

Title: Diverse definitions of prolonged labour and its consequences with sometimes subsequent inappropriate treatment

Authors:

Astrid Nystedt (astrid.nystedt@miun.se)
Ingegerd Hildingsson (ingegerd.hildingsson@miun.se)

Version: 4 Date: 9 June 2014

Author's response to reviews: see over
Dear Editor

We have included a title page at the front of our manuscript file. And it contains the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

We have uploaded the figure 1 as separate figure files using the "upload" form on the submission system only, and deleted the figure from the manuscript file. The legend and title are a part of the manuscript file, given after the reference list.

Best regards
Astrid Nystedt